Corporate Innovation: Do Diverse Boards Help?

Heng An, Carl R. Chen[®], Qun Wu, and Ting Zhang*

Abstract

We find that corporate innovation is positively related to board diversity as measured by a multidimensional index. The benefit of board diversity is more pronounced for firms with more complex operations, more experienced boards, and stronger external governance, suggesting that diverse boards have superior advising capacity. We find evidence to suggest that firms with diverse boards engage in more exploratory innovations and develop new technology in unfamiliar areas. As a result, they create a larger number of both most-cited and uncited patents. Finally, of the six different aspects of board diversity, professional diversity matters the most for corporate innovation.

I. Introduction

The composition of the board of directors has become a high-profile corporate governance issue, attracting substantial interest from shareholders, governments, and the media. Proponents of board diversity argue that it improves firm performance by enhancing monitoring and advising effectiveness. However, the push for diverse boards could be driven by social justifications rather than firm performance. Moreover, the effects of board diversity on firm performance have received limited research attention, with mixed evidence.¹ In this article, we focus

^{*}An, h_an@uncg.edu, University of North Carolina Bryan School of Business and Economics; Chen (corresponding author), rchen1@udayton.edu, and Zhang, tzhang1@udayton.edu, University of Dayton School of Business; and Wu, qunw@unr.edu, University of Nevada, Reno, College of Business Administration. We thank an anonymous referee, Shibo Bian, Ching-Hung Chang, Rui Lu, Paul Malatesta (the editor), Onur K. Tosun, Tong Zhou, and seminar participants at the 2016 Central Taiwan Finance Association Conference, the 2016 Financial Management Association (FMA) Meeting, the 2017 Asian Finance Association Conference, the 2017 FMA European Conference, the 2018 China Finance Review International Conference, Hunan University, National Chengchi University, National Dong-Hua University, Peking University, Renmin University of China, Southwest University of Finance and Economics, Sun Yet-San University, Tsinghua University, Xi'an Jiaotong University, Zhejiang University, and Zhejiang University of Finance and Economics for their helpful comments and suggestions. An acknowledges the financial support from the Kathleen Price Bryan Dean's Notable Scholar program. Wu thanks the University of Nevada, Reno, for summer research support, and Zhang thanks the University of Business Dean's research grant.

¹The empirical research disproportionately focuses on demographic diversity (e.g., Carter, D'Souza, Simkins, and Simpson (2010), Garcia-Meca, Garcia-Sanchez, and Martinez-Ferrero (2015), Upadhyay and Zeng (2014)), particularly gender diversity (e.g., Adams and Ferreira (2009),

on the potential impact of board diversity on innovation, a key indicator of firm performance.

Innovation can be a catalyst for growth and success of today's corporations (Porter (1992), Hall, Jaffe, and Trajtenberg (2005)). The board could play a vital role in shaping corporate innovation strategies. Charged with advising and monitoring managers, the board not only oversees a firm's performance but also actively contributes to strategic decisions. The resource-based view suggests that corporate innovation might benefit from the strategic resources brought by directors with different socioeconomic backgrounds. A diverse group of directors brings a variety of talents, expertise, and perspectives, which helps the firm and executives identify opportunities, generate ideas, and overcome blind spots.

According to a *Wall Street Journal* article, "diversity leads to more innovation, more outside-the-box thinking and better governance; when directors are too alike... they look at problems and solutions the same way" (Manzoni, Strebel, and Barsoux (2010)). Watson, Kumar, and Michaelsen (1993) and Wiersema and Bantel (1992) show that more diverse groups produce a greater range of perspectives and solutions to problems. Giannetti, Liao, and Yu (2015) find that directors with foreign experience bring brain gains to firms in emerging markets by transmitting knowledge and good management practices. Therefore, board diversity can be a competitive advantage for innovative firms. Nonetheless, pursuing board diversity is not without costs. Differences among directors can create communication and coordination problems in the boardroom, protracting the decision-making process. In short, diversity affects the workings of corporate boards.² The costbenefit tradeoff of board diversity indicates that its impact on corporate innovation is ultimately an empirical issue, which we address in this article.

To measure board diversity, we construct a multidimensional index based on six aspects of director characteristics: education, demographic attributes, cultural attitudes, managerial traits, professional backgrounds, and board experience. Using a large panel of U.S. public firms, we find a positive relation between board diversity and corporate innovation. Specifically, firms with a diverse board create a greater number of patents, and these patents receive significantly more citations. Besides patent and citation counts, we compare the innovation search strategies undertaken by firms with different board compositions. We find evidence to suggest that firms with diverse boards engage in more exploratory innovations,

Adams and Funk (2012)). Ferreira (2010) reviews the literature on board gender diversity and the role of women in the boardroom. Anderson, Reeb, Upadhyay, and Zhao (2011) and Chen and Zhang (2014) are two exceptions, who use a broader definition of diversity.

²Research on group diversity originates in the field of social psychology (see Levine and Moreland (1994) for a survey on group processes). When group members with different backgrounds work together, the effects can be combined in an additive fashion (e.g., Tuckman (1964)), combined in an assembly fashion (e.g., Rosenberg, Erlick, and Berkowitz (1955)), or dependent on the nature of the task (e.g., Steiner (1972)). In the management literature, Hambrick and Mason (1984) propose their upper echelon theory based on theories in social psychology (e.g., group dynamics) and strategic management (e.g., resource-based view), which has sparked intense research interest in top management team diversity. More recent empirical papers in management include Nielsen and Huse (2010) who find the proportion of female directors on the board is positively related to strategic control for a sample of 201 Norwegian firms, and Triana, Miller, and Trzebiatowski (2014) who find the relation between board gender diversity and corporate strategic change could be positive or negative depending on firm performance and the power of female directors.

developing new technology in unfamiliar areas. The pursuit of innovation in unknown areas can be risky since it increases the chances of both breakthroughs and failures. We find this to be true when analyzing the distribution of patent citations. As a board becomes more diverse, the probability distribution of patent quality flattens, ending up with long tails. That is, compared to firms with homogeneous boards, firms with diverse boards generate a larger number of both most-cited and uncited patents. However, there is no significant difference in the middle of the distributions. In other words, board diversity has little impact on incremental innovations. These findings suggest that a diverse board fosters a culture of innovation and creativity, which encourages thinking outside the box and venturing into novel technologies.

Although board diversity and innovation are positively correlated, their relation might not always be linear, given the cost–benefit tradeoff of board diversity. Piecewise regressions reveal that the relation between board diversity and innovations becomes insignificant when board diversity reaches very high levels, implying that the benefits of board diversity are offset by its costs. Therefore, policy makers and shareholders should exercise caution in designing the composition of a board. Diversity can be too much of a good thing.

The balance between the benefits and costs of board diversity can also vary across firms, depending on firms' internal and external characteristics. Specifically, we find the positive relation between board diversity and corporate innovation is more pronounced for firms subject to stronger external governance, such as institutional monitoring, external takeover threat, and market competitive pressure. The two main functions of a board are monitoring and advising managers. Strong external governance disciplines managers, rendering board monitoring less imperative. Therefore, the result that corporate innovation benefits more from board diversity when external governance is stronger suggests that the benefits are mainly due to the advising function of diverse boards. Additional cross-sectional tests show the superior advisory capacity of diverse boards is especially beneficial for firms with complex operations, operating in highly innovative industries, and with more experienced directors.

Finally, we assess which aspects of board diversity contribute to corporate innovation. Of the six diversity subindexes, we find professional diversity is the most important for innovation output. Overall, these results indicate that board diversity fosters corporate innovation. The results withstand a battery of robustness checks, including the instrumental variable (IV) approach, alternative measures of innovation, and controlling for firm and chief executive officer (CEO) fixed effects and several CEO characteristics related to innovation.

This article is related to several strands of the literature. First, our findings shed light on the debate over the benefits of board diversity. Despite the oftenstated belief that board diversity creates shareholder value, supporting evidence based on a comprehensive data set is still in short supply. A notable exception is a study by Anderson et al. (2011), who find a positive impact of board diversity on firm value. What is left unanswered, however, is the process by which board diversity creates value. By constructing one of the most comprehensive data sets on board diversity, we are able to uncover a mechanism by which board diversity affects shareholder value: by enhancing corporate innovation. After assessing six different aspects of board diversity, we find that professional diversity matters the most for corporate innovation.³

Second, and more broadly, our article is related to the recent literature that emphasizes the dual roles of boards as monitors and advisors of management. Empirical research focuses disproportionately on the monitoring role rather than the advising role of the board. The main source of director heterogeneity that most economists consider is whether directors are independent from managers. However, various dimensions of board diversity clearly become important when directors are viewed as resource providers. Our findings suggest that directors perform multiple functions and that innovation particularly benefits from the beneficial advisory capacity of diverse boards.

Third, our article contributes to the growing literature on the determinants of corporate innovation, especially those focusing on the importance of managerial and director traits.⁴ Using different measures of CEO overconfidence, Galasso and Simcoe (2011) and Hirshleifer, Low, and Teoh (2012) find that overconfident CEOs are better innovators. Islam and Zein (2018) show that CEOs' hands-on innovation experience enhances both the quality and quantities of their firms' innovation. Sunder, Sunder, and Zhang (2017) report that CEOs with pilot licenses are successful in corporate innovation. Faleye, Kovacs, and Venkateswaran (2014) find that better-connected CEOs invest more in R&D and their firms file more and higher-quality patents. Custodio, Ferreira, and Matos (2019) show that CEOs with general managerial ability spur innovation because they acquire knowledge beyond the firm's current technological domain and are less sensitive to the risk of termination. Baranchuk, Kieschnick, and Moussawi (2014) find that managers are motivated to pursue innovation through "innovation-friendly" incentive schemes with more deferred compensation, longer vesting periods for unexercised options, and some protection from takeover threats. Our evidence indicates that board diversity in general, and directors' professional diversity in particular, represents an important determinant of firm innovative activities.

Two recent papers are the most relevant to our article. Balsmeier, Fleming, and Manso (2017) report that board independence increases both patent and

³There is a growing literature in management that examines the relation between board diversity and corporate innovation. For example, Torchia, Calabro, and Huse (2011) find that at least three female directors are needed for gender diversity to enhance firm innovation for a sample of 317 Norwegian firms. Galia and Zenou (2012) examine the relation between board diversity and four types of innovation: product, process, organization, and marketing using survey data of 176 French firms in 2008. They find gender diversity is positively (negatively) related to marketing (product) innovation, while age diversity is positively (negatively) related to product (organization) innovation. Midavaine, Dolfsma, and Aalbers (2016) find that tenure (gender and education) diversity of directors is negatively (positively) correlated to the research and development (R&D) of 25 large U.S. companies.

⁴The determinants of corporate innovation identified in the literature include market competition (Aghion, Bloom, Blundell, and Griffith (2005)), antitakeover provisions (e.g., Meulbroek, Litchell, Mulherin, and Netter (1990), Atanassov (2013), Chemmanur and Tian (2018)), financial analysts coverage (e.g., He and Tian (2013)), business group affiliation (Belenzon and Berkovitz (2010)), local banking competition (Cornaggia, Mao, Tian, and Wolfe (2015)), firm alliances (Schilling and Phelps (2007)), ownership structure (e.g., Aghion, Van Reenen, and Zingales (2013), Bernstein (2015)), organizational structure (e.g., Seru (2014)), and the backing of private equity (Lerner, Sorensen, and Strömberg (2011)) and venture capital (Chemmanur, Loutskina, and Tian (2014)). This is by no means a complete list of the literature on the determinants of corporate innovation.

citation counts due to increased monitoring by independent directors. Chen, Leung, and Evans (2018) conclude that firms with more female directors obtain more patents and citations. Similarly, the authors attribute their findings to increased monitoring by female directors. While the two papers each focus on the level of a particular board characteristic and board monitoring, we examine the dispersion of a variety of board characteristics across directors (i.e., the second moment of the distribution of these characteristics). Importantly, after controlling for the level of board characteristics, we show that the diversity of board characteristics contributes to corporate innovation, which is mainly achieved through the superior advising capacity of diverse boards.

The remainder of this article is organized as follows: Section II describes the sample and variables. Section III presents the main empirical results and elaborates on the directors' advising role. Section IV discusses the innovation strategies. Section V conducts robustness checks, and Section VI concludes the article.

II. Sample and Variables

We extract board information and director biographies from MSCI GMI Ratings (formerly the Corporate Library), which covers more than 4,000 public firms in the United States. MSCI GMI Ratings provides annual information on board size and directors' age, gender, tenure, independence, executive positions, funder status, and external board seats. The database also provides a descriptive biography of each director, from which we extract the information on a director's nationality, education, professional experience, political connections, and military experience using SAS Text Miner.⁵ Our board data are from 2001 to 2007, consisting of 179,399 directorships (director-firm-year observations) in 15,672 firm-years, for a total of 4,448 firms and 43,639 directors.⁶

We obtain patent and citation data from three sources: the National Bureau of Economic Research (NBER) patent citation database; the data set collected by Kogan, Papanikolaou, Seru, and Stoffman (2017); and a Google bulk download. We obtain accounting data from the Compustat Fundamental File, stock price data from the Center for Research in Security Prices (CRSP), and institutional ownership data from the Thomson Reuters Institutional Holdings (13F) Database. We follow Hirshleifer et al. (2012) by excluding the industries (at the

⁵SAS Text Miner is a software program that discovers and analyzes unstructured text information. We primarily use the Text parsing node, the Text filter node, and the Text topic and text clustering nodes to extract the relevant information. The Text topic and text clustering nodes differ from traditional keyword searches, since they are able to identify and cluster *relevant* words for the same topic. See SAS's website (http://www.sas.com/text-analytics/text-miner/index.html) and the SAS Text Miner manual for more information.

⁶MSCI GMI Ratings no longer provides the detailed biography of each director necessary to construct our diversity measures for more recent years. Our main results hold using an alternative board diversity index based on Institutional Shareholder Services' (ISS) Director Data for the sample period up to 2012. Compared to The Corporate Library, however, the ISS Director Data lacks key data such as directors' nationalities, educational backgrounds/degrees, political connections, military experience, and business-founding experience. Therefore, this alternative index consists of four subindexes with 10 components, compared with six subindexes with 20 components for the original index. The regression results of using this alternative diversity index are not tabulated to save space but are available from the authors upon request.

2-digit Standard Industrial Classification level) without patenting activities since 1976. After combining the data sets, our final sample consists of 10,850 firm-year observations from 2001 to 2007.

A. Measures of Board Diversity

We construct the composite board diversity index based on six subindexes: i) a demographic diversity index, ii) an educational diversity index, iii) a professional diversity index, iv) a director experience index, v) a managerial trait index, and vi) a cultural diversity index. These subindexes are based on 20 components: age, gender, nationality, education degree, education background, expertise, business funding experience, executive position, tenure, external board seats, independence, board size, political connection, military experience, generation of Depression babies, power distance, individualism, uncertainty avoidance, masculinity, and long-term orientation.⁷

We first calculate a firm's index components in a year using the formulas in the Supplementary Material. We then rank each firm component based on sample quintiles of all firms in the year. For example, the age component is measured by the coefficient of variation (CV) of all directors' ages across the board. Because a higher value of this component indicates a higher level of diversity, the firm would earn a rank of 5 on age diversity if the value of its age component fell in the fifth quintile of all firms in that year.

To construct a subindex, we first sum the rank numbers of its corresponding components and then scale the sum by 100.⁸ Finally, the composite board diversity index is the sum of the six subindex scores. The Supplementary Material explains the construction of each subindex and the composite index in detail.

B. Measures of Innovation

We measure innovation output using patent-based metrics, which are better than input measures of innovation such as R&D (see He and Tian (2018) for the related discussions). We obtain information about all utility patents granted by the U.S. Patent and Trademark Office (USPTO) over the period of 1976–2006 from the NBER patent citation database. We supplement the patent and citation information from 2007 to 2010 with the Kogan et al. (2017) data set and from 2011 to 2015 using a Google bulk download.

We construct six measures of a firm's annual innovation output based on the patent application year.⁹ PATENT is the natural logarithm of 1 plus the firm's total number of patent applications filed in a year that are eventually granted. To correct for truncation, we construct the class-adjusted patent count by dividing

⁷We consider five cultural dimensions based on a director's nationality: power distance, individualism, uncertainty avoidance, masculinity, and long-term orientation (Hofstede (1984)). We obtained the cultural scores in 2013 from http://www.clearlycultural.com, which provides the Hofstede cultural index scores for 50 countries. If a director is from a country not on the list (less than 1% of all directors), we use a neighboring country with similar language, religion, and economic conditions. We explain the meaning of the five cultural measures in the Supplementary Material.

⁸Because the maximum rank sum of the 20 components is $100 (5 \times 20 = 100)$, we scale the indexes by 100 to make the composite diversity index range between 0 and 1.

⁹We use the application year instead of the grant year because the application year is closer to the actual time of innovation (Hall, Jaffe, and Trajtenberg (2001)).

a patent by the average number of patents of all firms in the same technology class for that year (see Hall et al. (2001), Seru (2014)). Therefore, our second measure, ADJ_PATENT, is the natural logarithm of 1 plus a firm's total classadjusted patent count in a year. To measure the quality of innovation, we use a patent's citation count, which is related to the social and economic value it has created. The citation count is also subject to truncation bias because it takes time for a patent to accumulate citations. To mitigate the truncation bias, we adjust the citation count by scaling the number of citations of a given patent by the average number of citations received by all patents in the same technology class that year. Therefore, our third measure, CITATION, is the natural logarithm of 1 plus a firm's total class-adjusted citation count in a year. Our fourth measure, ADJ_MV, is the natural logarithm of 1 plus the adjusted market value of the firms' new patents. Following Kogan et al. (2017), a patent's market value is calculated as the firm's market capitalization right before the patent announcement multiplied by the firm's market-adjusted stock return over the 3-day window around the announcement. Then, a patent's adjusted market value is the ratio of a patent's market value over the average market value of all patents in the same technology class that year. The fifth measure, AVG_EXT_CITATION, is the natural logarithm of 1 plus the average number of non-self-citations per patent received by the firm's patents in a year. Finally, ADJ_AVG_EXT_CITATION is the natural logarithm of 1 plus the firm's average external citation count scaled by the average number of external citations of all patents in the same technology class in that year. We follow Hall et al. (2001) by including both industry and year fixed effects in the regressions to address any remaining truncation bias.

C. Control Variables

First, we control for the levels of various board characteristics in the regressions to ensure our results are driven by the dispersion of characteristics across board members (i.e., the second moment of the distribution of these characteristics) rather than by their levels (i.e., the first moment of the distribution in these characteristics).¹⁰ Specifically, we control for the average age of the board members, the proportion of female directors on the board, the proportion of board directors with PhD degrees, the proportion of directors with business-founding experience, the average tenure of directors, the average number of external board seats held by the directors, the proportion of independent directors with military experience, the proportion of directors who were born during the Great Depression, and the directors' average cultural scores (power distance, individualism, uncertainty avoidance, masculinity, and long-term orientation).

Second, we control for the following firm characteristics that could affect a firm's innovation output: firm age, size, leverage, R&D, profitability, tangibility, cash flow, cash holding, Tobin's Q, firm risk, stock liquidity, and institutional ownership. The Appendix provides detailed variable definitions.

¹⁰We thank the referee for making this point.

D. Summary Statistics

Table 1 presents summary statistics. To mitigate the effects of outliers, we winsorize all continuous variables at the 1st and 99th percentiles. Panel A shows the statistics of board diversity variables. The mean (median) of the board diversity index is 0.605 (0.600) and the standard deviation is 0.060. Among the six subindexes, board experience diversity has the largest standard deviation (0.029), followed by cultural diversity (0.025), demographic diversity (0.023), and professional diversity (0.022). Panels B, C, and D present summary statistics for the innovation measures, board characteristics, and firm-level control variables, respectively. Finally, Panel E reports the annual average scores of board diversity index.

III. Empirical Results

A. Board Diversity and Innovation Output

We use the following baseline regression to examine the relation between board diversity and innovation output:

(1) INNOVATION_{*i*,*t*+3} =
$$\alpha + \beta_1 (\text{BRD}_D\text{IVERSITY})_{i,t}$$

+ $\sum_{j=2}^n \beta_j (\text{Control})_{i,t} + \mu_{i,t}$,

where the dependent variable is firm *i*'s annual innovation output, as measured by PATENT in year t + 3. Measuring innovation output in year t + 3 reflects the long-term nature of the innovation process. On the right side of equation (1), the main variable of interest is BRD_DIVERSITY, the board diversity index. We follow Hall et al. (2001) by including both year and industry fixed effects to mitigate the truncation bias.¹¹ We cluster standard errors at the firm level in all regressions.

Table 2 reports the regression results. The independent variables include BRD_DIVERSITY as well as the industry and year dummies in column 1. We add additional variables for board characteristics in column 2, firm-level characteristics in column 3, and both board and firm characteristics in columns 4 and 5. We replace the industry and year dummies by their interactions in column 4. The firm fixed effect model is not used due to a lack of within-firm variation on board diversity.

We find that the estimated coefficient of BRD_DIVERSITY is significantly positive across all five columns. In terms of statistical significance, BRD_DIVERSITY is significant at the 5% level (*t*-statistic = 2.43) in column 5 with the full set of control variables. In terms of economic significance, a 1-standard-deviation improvement in BRD_DIVERSITY will increase the value of PATENT by 0.049 (=0.810 × 0.060), which represents 5.73% of its mean.

We find that the level of many board characteristics is significantly related to corporate innovation. As column 5 shows, corporate innovation is negatively related to the proportion of directors born during the Great Depression but positively related to the number of external board seats and the

¹¹We classify industries by the 2-digit SIC code.

TABLE 1 Summary Statistics

Table 1 reports the summary statistics of board diversity indexes (Panel A), innovation measures (Panel B), board characteristics (Panel C), and firm characteristics (Panel D). Panel E reports the annual average scores of board diversity indexes. Variables in Panel A and E are defined in the Supplementary Material and those in Panel B–D are defined in the Appendix.

Variable	Mean	Std. Dev.	Min.	P25	Median	P75	Max
Panel A. Board Diversity Indexes							
BRD_DIVERSITY DEMOGRAPHIC_DIVERSITY EDUCATIONAL_DIVERSITY PROFESSIONAL_DIVERSITY EXPERIENCE_DIVERSITY MANAGERIAL_TRAIT_DIVERSITY CULTURAL_DIVERSITY	0.605 0.089 0.062 0.090 0.150 0.060 0.154	0.060 0.023 0.019 0.022 0.029 0.013 0.025	0.400 0.030 0.020 0.040 0.070 0.050 0.150	0.567 0.070 0.050 0.070 0.130 0.050 0.150	0.600 0.090 0.050 0.090 0.150 0.050 0.150	0.644 0.100 0.080 0.110 0.170 0.070 0.150	0.867 0.150 0.100 0.150 0.250 0.100 0.250
Panel B. Innovation Measures							
PATENT ADJ_PATENT CITATION ADJ_MV AVG_EXT_CITATION ADJ_AVG_EXT_CITATION UNKNOWN TECH_PROXY TOP1_CITE TOP10_CITE MIDDLE_CITE NO_CITE	0.855 0.587 0.159 0.527 0.149 0.148 0.235 0.876 0.876 0.137 0.321 0.649 0.475	1.425 1.498 0.589 1.359 0.391 0.358 0.531 0.254 0.461 0.810 1.274 1.093	0.000 0.000 0.000 0.000 0.000 0.000 0.000 0.000 0.000 0.000 0.000 0.000	0.000 0.000 0.000 0.000 0.000 0.000 0.000 0.000 0.000 0.000 0.000 0.000	0.000 0.000 0.000 0.000 0.000 0.000 1.000 0.000 0.000 0.000 0.000	1.113 0.455 0.000 0.306 0.000 0.000 1.000 0.000 0.000 0.000 0.698 0.000	8.357 10.824 6.642 8.561 3.807 3.324 3.973 1.000 4.127 6.021 8.168 7.423
Panel C. Board Characteristics							
BRD_AGE BRD_FEMALE BRD_FOUNDER BRD_TENURE BRD_TENURE BRD_EXT_SEAT BRD_INDEPENDENCE BRD_POLITICAL_CONNECTION BRD_MILITARY BRD_DEPRESSION_BABY POWER_DISTANCE INDIVIDUALISM UNCERTAINTY_AVOIDANCE MASCULINITY LONG_TERM_ORIENTATION Panel D_Firm Characteristics	4.085 0.092 0.067 0.024 2.130 1.035 0.766 0.008 0.004 0.000 0.400 0.907 0.461 0.620 0.291	0.073 0.087 0.111 0.060 0.447 0.275 0.116 0.030 0.024 0.100 0.010 0.021 0.015 0.012 0.016	3.728 0.000 0.000 0.000 0.182 0.000 0.000 0.000 0.000 0.130 0.170 0.230 0.160 0.280	4.043 0.000 0.000 1.863 0.821 0.700 0.000 0.000 0.000 0.400 0.910 0.460 0.620 0.290	4.092 0.091 0.000 2.169 1.022 0.786 0.000 0.000 0.000 0.400 0.400 0.910 0.460 0.620 0.290	4.134 0.143 0.105 0.000 2.431 1.237 0.857 0.000 0.000 0.100 0.400 0.910 0.460 0.620 0.290	4.323 0.667 0.857 0.500 3.455 2.005 1.000 0.375 0.500 0.704 0.680 0.910 0.860 0.785 0.870
SIZE AGE LEV ROA CASH CF TOBINS_Q TANGIBLE R&D TURNOVER RISK IO	7.495 2.714 0.554 0.114 0.096 0.028 0.225 0.030 1.674 0.110 0.615	1.670 0.896 0.246 0.113 0.115 0.111 0.240 0.226 0.060 1.437 0.064 0.256	4.167 0.000 0.072 -0.378 0.000 -0.583 0.104 0.000 0.000 0.092 0.027 0.000	6.298 2.197 0.374 0.053 0.018 0.010 0.392 0.043 0.000 0.748 0.062 0.364	7.363 2.713 0.552 0.107 0.051 0.038 0.563 0.150 0.000 1.234 0.090 0.670	8.528 3.427 0.731 0.165 0.131 0.077 0.737 0.334 0.033 2.085 0.131 0.813	12.183 4.382 0.976 0.410 0.565 0.243 1.168 0.878 0.335 8.139 0.364 1.000
Panel E. Average Board Diversity Sc	ores by Yea	r					
Variable	2001	2002	2003	2004	2005	2006	2007
BRD_DIVERSITY DEMOGRAPHIC_DIVERSITY EDUCATIONAL_DIVERSITY PROFESSIONAL_DIVERSITY EXPERIENCE_DIVERSITY MANAGERIAL_TRAIT_DIVERSITY CULTURAL_DIVERSITY No. of obs	0.601 0.088 0.058 0.091 0.153 0.061 0.150	0.607 0.092 0.059 0.094 0.149 0.063 0.150	0.616 0.090 0.061 0.094 0.148 0.063 0.159	0.609 0.087 0.062 0.093 0.153 0.059 0.156 1.372	0.609 0.087 0.063 0.0933 0.151 0.058 0.157 1.546	0.601 0.089 0.063 0.087 0.148 0.060 0.155 2.396	0.599 0.088 0.063 0.084 0.149 0.060 0.154 2.426
110. 01 0003.	1,011	1,000	1,001	1,012	1,040	2,000	0_++2

TABLE 2 Relation between Board Diversity and Corporate Innovation

Table 2 reports the results of the baseline regression in equation (1). The dependent variable is corporate innovation as measured by PATENT. The main variable of interest is BRD_DIVERSITY, the composite board diversity index. The *t*-statistics (in parentheses) are based on firm-level clustered standard errors. The table also presents the number of observations and R^2 values for each regression. *, **, and *** indicate statistical significance at the 10%, 5%, and 1% levels, respectively. All other variables are defined in the Appendix.

			PATENT _{$t+3$}		
Dependent Variable	1	2	3	4	5
BRD_DIVERSITY	2.737*** (6.62)	1.953*** (4.79)	0.963*** (3.14)	0.828** (2.43)	0.810** (2.43)
BRD_AGE		1.253*** (3.49)		0.026 (0.08)	0.040 (0.12)
BRD_FEMALE		1.248*** (4.80)		0.460** (2.05)	0.530** (2.35)
BRD_PHD		1.252*** (5.18)		0.765*** (3.70)	0.753*** (3.58)
BRD_FOUNDER		0.908** (2.33)		0.986*** (3.16)	1.034*** (3.33)
BRD_TENURE		-0.115 (-1.22)		-0.118 (-1.25)	-0.114 (-1.02)
BRD_EXT_SEAT		1.092*** (10.90)		0.315*** (3.62)	0.352*** (4.13)
BRD_INDEPENDENCE		0.771*** (4.30)		0.182 (1.13)	0.277* (1.77)
BRD_POLITICAL_CONNECTION		1.222* (1.68)		1.010 (1.51)	0.726 (1.31)
BRD_MILITARY		-1.269 (1.31)		0.647 (0.73)	-1.018 (-1.14)
BRD_DEPRESSION_BABY		-1.351*** (-5.01)		-0.591** (-2.32)	-0.546** (-2.21)
POWER_DISTANCE		2.480 (0.75)		1.773 (0.61)	1.403 (0.50)
INDIVIDUALISM		1.940 (1.13)		0.170 (0.11)	0.308
UNCERTAINTY_AVOIDANCE		1.365 (0.76)		1.551 (0.98)	1.893 (1.24)
MASCULINITY		-0.195 (-0.08)		-0.746 (-0.35)	-0.605 (-0.29)
LONG_TERM_ORIENTATION		1.013 (0.36)		1.349 (0.58)	1.140 (0.51)
SIZE			0.399*** (16.24)	0.370*** (14.39)	0.364*** (14.28)
AGE			0.113*** (4.39)	0.173*** (5.52)	0.169*** (5.36)
LEV			-1.372*** (-3.90)	-1.352*** (-3.81)	-1.284*** (-3.66)
ROA			1.721***	1.674*** (6.20)	1.769***
CASH			0.038	0.036	0.013
CF			-0.255	-0.172 (-0.76)	-0.247 (-1.09)
TOBINS_Q			0.987*** (2.87)	0.898***	0.849***
TANGIBLE			0.043	0.035	0.025
R&D			6.428*** (12.31)	6.186*** (11.84)	5.709*** (11.22)
				(continued	on next page)

			ΡΔΤΕΝΤ -		
Dependent Variable	1	2	3	4	5
TURNOVER			0.043 (1.60)	0.041 (1.58)	0.040 (1.50)
RISK			0.544* (1.74)	0.212 (0.66)	0.411 (1.34)
Ю			0.180* (1.68)	0.187* (1.66)	0.181* (1.65)
Industry effects Year effects Industry-year effects	Yes Yes No	Yes Yes No	Yes Yes No	 Yes	Yes Yes No
No. of obs.	10,850	10,850	10,850	10,850	10,850
R^2	0.332	0.406	0.508	0.557	0.524

TABLE 2 (continued) Relation between Board Diversity and Corporate Innovation

proportion of female directors, independent directors, directors with PhD degrees, and directors with business-founding experience. As for the economic significance, while a 1-standard-deviation increase in BRD_DEPRESSION_BABY will decrease the value of PATENT by 0.055 (= -0.546×0.100), or 6.43% of its mean, a 1-standard-deviation increase in BRD_FEMALE, BRD_PHD, BRD_FOUNDER, BRD_EXT_SEAT, and BRD_INDEPENDENCE will increase the value of PATENT by 0.046 (= 0.530×0.087), 0.084 (= 0.753×0.111), 0.062 (= 1.034×0.060), 0.097 (= 0.352×0.275), and 0.032 (= 0.277×0.116), representing 5.38%, 9.82%, 7.25%, 11.35%, and 3.74% of the mean of PATENT, respectively.¹² Overall, the results here suggest that corporate innovation is correlated to both the level and dispersion of various board characteristics, with comparable economic significance.

Among firm characteristics, we find innovation output to be negatively related to leverage but positively related to size, age, profitability, Tobin's Q, R&D expense ratio, and institutional ownership, consistent with prior studies (e.g., Fang, Tian, and Tice (2014)).

B. Nonlinear Relation between Innovation and Board Diversity

The results thus far show that the overall relation between board diversity and corporate innovation is positive, suggesting that diversity fosters innovation. However, the pursuit of board diversity is not without costs. Differences among directors can create communication barriers and coordination problems in the board-room, protracting the decision-making process. The cost–benefit tradeoff of board diversity indicates that its relation with corporate innovation could be nonlinear. To examine the potential nonlinear relation between board diversity and firm innovation, we employ the following piecewise regression models, where 0.7111 is the 95th percentile of BRD_DIVERSITY:¹³

 $^{^{12}}$ To estimate the economic significance, we use the estimated coefficients in column 5 of Table 2 and the standard deviations in Panel C of Table 1.

¹³We use the SAS NLIN function to identify the cutoff point of BRD_DIVERSITY that best fits two adjoined lines in the piecewise regression.

(2) INNOVATION_{*i*,*t*+3} =
$$\alpha + \beta_1(\text{BRD}_\text{DIVERSITY} < 0.7111)_{i,t}$$

+ $\beta_2(\text{BRD}_\text{DIVERSITY} \ge 0.7111)_{i,t} + \sum_{j=3}^{n} \beta_j(\text{Control})_{i,t} + \mu_{i,t}$
BRD_DIVERSITY < 0.7111 =
 $\begin{cases} \text{BRD}_\text{DIVERSITY} & \text{If BRD}_\text{DIVERSITY} < 0.7111\\ 0.7111 & \text{If BRD}_\text{DIVERSITY} \ge 0.7111 \end{cases}$
BRD_DIVERSITY ≥ 0.7111 =
 $\begin{cases} 0 & \text{If BRD}_\text{DIVERSITY} < 0.7111\\ \text{BRD}_\text{DIVERSITY} - 0.7111 & \text{If BRD}_\text{DIVERSITY} < 0.7111 \end{cases}$

Table 3 reports the piecewise regression results. While the estimated coefficient of BRD_DIVERSITY < 0.7111 remains significantly positive, that of the variable BRD_DIVERSITY \geq 0.7111 is -0.143, with a *t*-statistic of -0.56. The regression results show that the relation between diversity and innovation outputs becomes insignificant after diversity reaches a very high level, indicating that the benefits of board diversity are offset by its costs.

C. Advising versus Monitoring: Cross-Sectional Tests

The balance between the benefits and costs of board diversity can also vary across firms. In this section, we employ cross-sectional tests to examine if the beneficial effects of board diversity vary in firms' internal characteristics and external

TABLE 3 Nonlinear Relation between Board Diversity and Innovation						
Table 3 reports the piecewise reg	ression resu	lts of e	equation (2):			
INNOVATION _{i,t+3}	$= \alpha + \beta_1(\theta)$	BRD_D	DIVERSITY < 0.7111) _{<i>i</i>,<i>t</i>}			
$+\beta_2(BRD_DIVER$	+ $\beta_2(\text{BRD}_\text{DIVERSITY} \ge 0.7111)_{i,t} + \sum_{i=3}^n \beta_i(\text{Control})_{i,t} + \mu_{i,t}$					
BRD_DIVERSITY	< 0.7111	=	BRD_DIVERSITY	If BRD_0 If BRD_0	DIVERSITY < 0.7111 DIVERSITY ≥ 0.7111	
BRD_DIVERSITY	≥ 0.7111	=	{0 BRD_DIVERSITY - 0	0.7111	If BRD_DIVERSITY < 0.7111 If BRD_DIVERSITY \geq 0.7111.	

The dependent variable is PATENT. The regression controls for both year and industry fixed effects. The *t*-statistics (in parentheses) are based on firm-level clustered standard errors. The table also presents the number of observations and the R^2 values for each regression. *, **, and *** indicate statistical significance at the 10%, 5%, and 1% levels, respectively.

Dependent Variable	PATENT _{t+3}
BRD_DIVERSITY < 0.7111	0.962*** (2.88)
BRD_DIVERSITY ≥ 0.7111	-0.143 (-0.56)
Control variables Industry effects Year effects	Yes Yes Yes
No. of obs. R^2	10,850 0.524

monitoring strength.¹⁴ The two main functions of the board are monitoring and advising managers. Strong external governance disciplines managers, rendering board monitoring less imperative and enabling directors to contribute more of their limited resources to advising. Therefore, we first split the sample based on the strength of external governance mechanism: independent institutional ownership, external takeover threat, and market competition. Table 4 summarizes the three cross-sectional test results.

1. Independent Institutional Ownership

Institutional monitoring has long been considered an important governance solution to the agency issue. Because not all institutional investors are effective monitors due to potential business ties with their portfolio firms (e.g., bankaffiliated institutions), we follow the literature by calculating the ownership of independent institutional investors, which include public pension funds, investment companies, and independent advisors (e.g., Brickley, Lease, and Smith (1988), Chen, Harford, and Li (2007), Cornett, Marcus, Saunders, and Tehranian (2007)). We split the sample based on independent institutional ownership (IIO) and assign an observation to the high-IIO (low-IIO) subsample if its IIO is in the top (bottom) tercile of all sample firms in a year. We run the baseline regression separately for the two subsamples, including all the control variables in column 5 of Table 2. Column 1 of Table 4 compares the results. The estimated coefficient of board diversity is positive and highly significant, with a t-statistic of 4.73 in the high-IIO subsample compared to a *t*-statistic of 1.69 in the low-IIO subsample. The χ^2 test shows that the estimated coefficients of board diversity are significantly different in the two subsamples, with a χ^2 statistic of 3.34 (*p*-value = 0.067). As independent institutional investors provide important external monitoring, rendering

TABLE 4 Cross-Sectional Tests: External Governance Strength

Table 4 compares the regression results for firms with strong versus weak external governance, as measured by independent institutional ownership (IIO) (column 1), takeover threat (column 2), and market competition (column 3). The dependent variable is corporate innovation as measured by PATENT. The main variable of interest is BRD_DIVERSITY, the composite board diversity index. The table also reports the results of the χ^2 tests on the difference in the coefficients on the diversity index between the subsamples. All regressions control for year and industry fixed effects. The *t*-statistics (in parentheses) are based on firm-level clustered standard errors. *, **, and *** indicate statistical significance at the 10%, 5%, and 1% levels, respectively.

			PATE	NT _{t+3}		
	IIC	1	Takeover	Threat	Market Co	ompetition
	1		2		3	3
Dependent Variable	High	Low	High	Low	High	Low
BRD_DIVERSITY	1.155*** (4.73)	0.707* (1.69)	0.932*** (3.91)	0.542 (1.43)	0.947*** (3.79)	0.601 (0.60)
Control variables Industry effects Year effects	Yes Yes Yes	Yes Yes Yes	Yes Yes Yes	Yes Yes Yes	Yes Yes Yes	Yes Yes Yes
No. of obs. R ²	3,611 0.572	3,599 0.601	3,677 0.537	3,592 0.533	3,650 0.584	3,582 0.531
Difference in coeff. χ^2 statistic (<i>p</i> -value)	0.448* 3.34 (0.067)		0.390* 2.84 (0.092)		0.346* 3.71 (0.054)	

¹⁴We thank the referee for suggesting many cross-sectional tests in this section.

board monitoring capacity less imperative, the results reported here suggest that the beneficial impact of a diverse board on innovation is largely due to the board's advising function.

2. External Takeover Threat

Next, we split the sample based on external takeover threat, another external governance mechanism (Haan and Riyanto (2006)). Using principal component analysis, we calculate a takeover threat score for each firm-year observation based on the following three takeover threat measures used in the literature: an industry-specific takeover probability calculated as the relative frequency of takeovers in a firm's industry over the past 3 years (Agrawal and Knoeber (1998)), a dummy variable that equals 1 if the firm is acquired in the 7-year period after the current year in question (Agrawal and Knoeber (1998)), and a state antitakeover index that is the count of five state statutes (the control share acquisition, fair price, business combination, poison pill endorsement, and constituency statutes) as in Bebchuk and Cohen (2003).¹⁵ We split the sample based on the takeover-threat score and assign an observation to the high-takeover-threat (low-takeover-threat) subsample if the takeover threat score is in the top (bottom) tercile of all sample firms in a year.

As column 2 of Table 4 shows, the estimated coefficient of board diversity is 0.932 and statistically significant, with a *t*-statistic of 3.91, in the high takeover threat subsample but is not significant in the low-takeover-threat subsample. The χ^2 test shows that the estimated coefficients of board diversity are significantly different in the two subsamples.

3. Market Competition

We now turn to the potential impact of product market competition on the link between board diversity and corporate innovation. Competitive pressure from the product market disciplines managers, rendering the board monitoring function less imperative. Giroud and Mueller (2011) show that firms in competitive industries benefit less from better corporate governance. To measure competition, we calculate the Herfindahl-Hirschman Index (HHI) of each industry as the sum of the squared market shares of all the Compustat firms in that industry. Higher values of HHI indicate greater market concentration and lower competition in the industry. We split the sample based on the HHI scores and assign an observation to the low-competition (high-competition) subsample if its HHI is in the top (bottom) tercile of all sample firms in a year. We run the baseline regression model separately for the two subsamples; column 3 of Table 4 compares the results. The estimated coefficient of board diversity is 0.947 and statistically significant in the high-competition subsample, with a *t*-statistic of 3.79, but is not significant in the low-competition subsample. The χ^2 test shows that the estimated coefficients of board diversity are significantly different in the two subsamples. Consistent with previous results on IIO and external takeover threat, the result that board diversity is only significantly related to innovation in competitive industries reinforces the

¹⁵Since a higher state takeover index indicates stronger legal protection against hostile takeovers (i.e., lower takeover threat), we use five minus the state takeover index for the principal component analysis.

notion that the beneficial impact of a diverse board on innovation is mainly due to the board's advising function.

Overall, the results in Table 4 show that corporate innovation benefits more from board diversity when external governance is stronger, which suggests that the advising function is the main channel through which board diversity enhances innovation. Next, we conduct three additional cross-section tests to examine which types of firms tend to benefit more from the superior advising function of a diverse board.

4. Board Experience

First, we split the sample based on director experience. We expect the beneficial effect of a diverse board to be more pronounced when the directors have more experience. We classify a board as experienced if at least one of its directors has filed any patent(s) and/or has business-founding experience. We split the sample into experienced and inexperienced director subsamples and run the baseline regressions separately for the two subsamples.¹⁶ As column 1 of Table 5 shows, the estimated coefficient of board diversity in the experienced subsample is 1.531, with a *t*-statistic of 4.70, and 0.980 in the inexperienced subsample, with a *t*-statistic of 2.95 (*p*-value = 0.091). The result that the impact of board diversity on innovation is more pronounced when directors are more experienced further confirms the superior advisory capacity of diverse boards.

5. Innovation-Intensive Industries

10%, 5%, and 1% levels, respectively.

Next, we split the sample firms based on the varying levels of innovative intensity across industries. We expect firms in innovation-intensive industries to

IABLE 5	
Cross-Sectional Tests: Advising Benefits	
Table 5 compares the regression results for firms with experienced versus inexperienced directors (column 1), o	cerating
in madstines with high versus low innovation (courner2), and with complex versus honcomplex operations (courner dependent variable is corrected in providion as massived by PATENT the main variable of intersect is RPD. DIV	EDSITV
dependent variable is corporate minoration as measured by rank the results of the x^2 tests on the difference in the cost	efficients
The diversity index between the subsamples. All regressions control for very and industry fixed effects. The t -s	tatistics
(in parentheses) are based on firm-level clustered standard errors. *, **, and *** indicate statistical significance	e at the

				PATENT _{t+3}		
	Director	Experience	Industry Ir	nnovation Intensity	Firm C	Complexity
Dependent Variable	Experienced	Inexperienced	High	Low	Complex	Noncomplex
BRD_DIVERSITY	1.531*** (4.70)	0.980* (1.90)	0.933*** (3.12)	0.712* (1.78)	1.305*** (4.55)	0.230 (0.64)
Control variables Industry effects Year effects	Yes Yes Yes	Yes Yes Yes	Yes Yes Yes	Yes Yes Yes	Yes Yes Yes	Yes Yes Yes
No. of obs. R^2	5,432 0.557	5,418 0.491	3,635 0.534	3,625 0.348	2,566 0.520	2,570 0.458
Difference in coeff. χ^2 statistic (<i>p</i> -value)	0 2 (0	.551* .95 .091)		0.221 1.75 (0.186)	(1.075** 4.18 0.041)

¹⁶We exclude BRD_FOUNDER from the regressions.

benefit more from the superior advisory function of diverse boards. We classify an industry as high (low) innovation if the average number of patents of all firms in the industry is in the top (bottom) tercile of all industries in a year.

As column 2 of Table 5 shows, the estimated coefficient of board diversity in high-innovation industries is 0.933, with a *t*-statistic of 3.12, and 0.712 in the low-innovation industries. However, the difference between the estimated coefficients of board diversity in the two subsamples is not statistically significant at conventional levels, with a χ^2 statistic of 1.75.

6. Firm Complexity

Finally, we split the sample based on the complexity of a firm's operations.¹⁷ Using principal component analysis, we calculate a complexity score based on six variables: the number of business segments, 1 minus the HHI based on business segment sales, the number of geographic segments, 1 minus the HHI based on geographic segment sales, total assets, and the number of employees. We assign an observation to the noncomplex (complex) subsample if its complexity score is in the bottom (top) tercile of all sample firms in a year.

As column 3 of Table 5 shows, the estimated coefficient of board diversity in the complex subsample is 1.305 and statistically significant, with a *t*-statistic of 4.55, and 0.230 in the noncomplex subsample, with a *t*-statistic of 0.64. The χ^2 test shows that the estimated coefficients of board diversity are significantly different in the two subsamples.¹⁸ The results suggest that the advisory function of the board is more important for firms with more complex operations.

D. Board Diversity Subindex

Our results so far are based on the board diversity index, which is constructed by combining six subindexes: i) the demographic diversity index, ii) the educational diversity index, iii) the professional diversity index, iv) the director experience index, v) the managerial trait index, and vi) the cultural diversity index. In this section, we regress PATENT against the six subindexes to examine which aspects of board diversity matter the most for corporate innovation. We include the same set of control variables in the regression as in column 5 of Table 2.

Table 6 reports the results. The estimated coefficients of educational diversity and professional diversity are positive and statistically significant at 10% and 1% respectively. As described in detail in the Supplementary Material, educational diversity is based on the variation in degrees and background; professional diversity is based on the variation in expertise, founding experience, and executive positions. The results suggest that the diverse educational backgrounds and professional experiences of board members contribute to the generation of patents. In addition to its statistical significance, the economic significance of professional diversity is sizable. A 1-standard-deviation increase in the professional diversity index will increase PATENT by about 0.051, a 5.94% increase in the mean. The estimated coefficients of director experience, managerial traits, and demographic

¹⁷The availability of Compustat Segment data reduces the sample size to 7,786.

¹⁸Our results are robust to the alternative measure of firm complexity defined as 1 minus the firm's HHI based on segment sales.

TABLE 6 Subindexes of Board Diversity

Table 6 reports the effects of six subindexes of board diversity (described in the Supplementary Material) on innovation. The dependent variable is corporate innovation as measured by PATENT. The regression controls for year and industry fixed effects. The *t*-statistics (in parentheses) are based on firm-level clustered standard errors. The table also presents the number of observations and the R^2 values for each regression. *, **, and *** indicate statistical significance at the 10%, 5%, and 1% levels, respectively.

Dependent Variable	
DEMOGRAPHIC_DIVERSITY	0.674 (0.73)
EDUCATIONAL_DIVERSITY	1.916* (1.94)
PROFESSIONAL_DIVERSITY	2.301*** (3.30)
EXPERIENCE_DIVERSITY	-0.466 (-0.74)
MANAGERIAL_TRAIT_DIVERSITY	-0.960 (-0.38)
CULTURAL_DIVERSITY	1.299 (1.52)
Control variables Industry effects Year effects	Yes Yes Yes
No. of obs. R ²	10,850 0.525

and cultural diversity are not statistically significant.¹⁹ To summarize, a board's professional diversity is the diversity subindex that matters the most for innovation productivity, and education diversity is the next most important.²⁰

IV. Board Diversity and Innovation Strategies

In this section, we construct indicators of patent quality to measure not only the number of innovative patents but also the strategic direction of innovation undertaken by firms with different board compositions. In particular, we test whether firms with diverse boards pursue exploratory or exploitative innovative strategies. If a diverse board encourages more exploration and thinking outside the box, we would expect these firms to venture into less familiar areas and develop new technologies. Alternatively, firms could choose exploitative strategies by relying on conventional technologies or staying in areas that are more familiar to them.

To test our conjectures, we first calculate the number of patents that a firm files in previously unknown technology classes. We follow Balsmeier et al. (2017) by defining unknown patent classes as USPTO technology classes in which the firm has not applied for any patent since 1976. The variable UNKNOWN is the natural logarithm of 1 plus the number of patents filed in previously unknown areas.

¹⁹The insignificant effect of demographic and cultural diversity here is consistent with Rhode and Packel (2014), who show that demographic and cultural diversity are driven by social justification rather than firm performance.

²⁰We also construct the first principal component of each subindex to examine how each of the six principal components relates to future innovation. We find the results based on the first principal components similar to those reported in Table 6.

Second, we calculate the technological proximity (TECH_PROXY) between the firm's patent portfolio in 1 year and that of the same firm in prior years. Specifically, we follow Balsmeier et al. (2017) and Jaffe (1989) by calculating technological proximity according to the following formula:

(3) TECH_PROXY_{*it+3*} =
$$\sum_{k=1}^{K} f_{ikt+3} f_{ikt+2} / \left(\sum_{k=1}^{K} f_{ikt+3}^2 \times \sum_{k=1}^{K} f_{ikt+2}^2 \right)^{1/2}$$
,

where f_{ikt+3} is the fraction of firm *i*'s patents that belong to technology class *k* in year t + 3 and f_{ikt+2} is the fraction of firm *i*'s patent portfolio up to year t + 2 that belongs to technology class k.²¹ The variable TECH_PROXY is continuous and ranges between 0 and 1. The higher the technological proximity, the more overlap between the technology class distributions of a firm's patent portfolios over time. For example, the value of 1 indicates that the technology class distribution of a firm's patent portfolio in a year is exactly the same as that of the previous years.

Next, we regress the two new variables separately against the board diversity index to test whether firms with different board compositions remain in or deviate from known research areas. All the independent variables are the same as in column 5 of Table 2. As Table 7 shows, the estimated coefficient of the board diversity index is positive and significant (*t*-statistic = 1.91) when the dependent variable is UNKNOWN, indicating that board diversity is positively related to the number of patents filed in previously unknown areas. We obtain similar results when using technological proximity as a more sophisticated measure of the corporate innovation trajectory. The estimated coefficient of the board diversity index is significantly negative, with a *t*-statistic of -1.90, when the dependent variable is TECH_PROXY. Overall, the results suggest that diverse boards encourage firms to explore novel technology and venture into previously unfamiliar areas.

The pursuit of innovation in unknown areas can be risky since it increases the chances of both breakthroughs and failures. Specifically, we expect explorative

TABLE 7 Board Diversity and Innovation Strategies

Dependent Variable		TECH_PROXY _{t+3}
BRD_DIVERSITY	0.270* (1.91)	-0.102* (-1.90)
Control variables Industry effects Year effects	Yes Yes Yes	Yes Yes Yes
No. of obs. R^2	10,850 0.249	10,850 0.152

²¹Our results are robust to measuring the firm's technological proximity between its patent portfolio in year t + 3 and that up to year t.

innovation strategies to flatten the probability distribution of innovation quality. In other words, as firms venture into less familiar areas and develop new technologies, they should generate greater numbers of both breakthroughs and marginal inventions, leading to a more flattened distribution with long tails. To measure the quality of innovation, we follow Balsmeier et al. (2017) by splitting all the patents in our sample into four mutually exclusive categories by the number of patent citations received. Breakthroughs are patents receiving a number of citations in the top 1% of the citation distribution within the technology class and application year. Important innovations are patents that fall in the top 10% of the citation distribution within the technology class and application year (excluding the top 1%). On the other end of the distribution, there are patents that receive no citations at all. The remaining patents are classified as incremental innovations, since they receive citations but are not in the top 10%. We use the logarithm of 1 plus the number of patents in each of the aforementioned categories as the dependent variables (i.e., TOP1_CITE, TOP10_CITE, MIDDLE_CITE, and NO_CITE) and regress each of them against the board diversity index. We use the same set of control variables as specified in column 5 of Table 2.

Table 8 reports the results. The estimated coefficients of the board diversity index are 0.726, 0.706, 0.649, and 0.702 when the dependent variables are TOP1_CITE, TOP10_CITE, MIDDLE_CITE, and NO_CITE, respectively. Moreover, the statistical significance of board diversity varies. Specifically, the *t*-statistics are 2.64, 2.41, 1.60, and 2.15 when the dependent variables are TOP1_CITE, TOP10_CITE, MIDDLE_CITE, and NO_CITE, respectively. The results indicate that firms with diverse boards generate slightly more breakthroughs and patents without citations. In addition to a positive relation between board diversity and innovation output as measured by the firm's patent count, we also find evidence to suggest that board diversity matters for innovation strategies. That is, the increase in patents is mainly due to the tails of the patent quality distribution (i.e., patents with the most citations and those without citations). The evidence is also consistent with earlier findings in this section, that firms with diverse boards

TABLE 8 Board Diversity and Patent Distribution

Table 8 reports the results of regressions regarding board diversity on the distribution of patent citations. The dependent variables are TOP1_CITE (the natural logarithm of 1 plus the number of patents in the top 1% of the citation distribution within the technology class and application year), TOP1_CITE (the natural logarithm of 1 plus the number of patents in the top 10% of the citation distribution within the technology class and application year, excluding the top 1%), MIDDLE_CITE (the natural logarithm of 1 plus the number of patents that are cited but are not in the top 10% of the citation distribution), and NO_CITE (the natural logarithm of 1 plus the number of patents that are cited but are not in the top 10% of the citation distribution), and NO_CITE (the natural logarithm of 1 plus the number of patents that are cited but are not in the top 10% of the citation distribution), and NO_CITE (the natural logarithm of 1 plus the number of patents that are cited but are not in the top 10% of the citation distribution). The second diversity index. All regressions control for both industry and year fixed effects. The *t*-statistics (in parentheses) are based on firm-level clustered standard errors. The table also presents the number of observations and the *R*² values for each regression. *, **, and *** indicate statistical significance at the 10%, 5%, and 1% levels, respectively.

Dependent Variable	TOP1_CITE _{t+3}	TOP10_CITE _{t+3}	MIDDLE_CITE _{t+3}	NO_CITE _{t+3}
BRD_DIVERSITY	0.726***	0.706**	0.649	0.702**
	(2.64)	(2.41)	(1.60)	(2.15)
Control variables	Yes	Yes	Yes	Yes
Industry effects	Yes	Yes	Yes	Yes
Year effects	Yes	Yes	Yes	Yes
No. of obs. R^2	10,850	10,850	10,850	10,850
	0.212	0.316	0.412	0.372

adopt an exploration strategy by venturing into unfamiliar areas, resulting in a larger number of both most-cited and uncited patents. Taken together, the results in this section suggest that a diverse board fosters a culture of innovation and creativity, which encourages thinking outside the box and the development of novel technologies.

V. Robustness Checks

A. Instrumental Variable Estimation

The results thus far have been consistent with the notion that board diversity fosters corporate innovation. Nonetheless, we recognize that the potential endogeneity of board diversity makes causality difficult to identify. One source of concern is reverse causality. For example, board diversity could be affected by corporate innovation if more innovative firms attract directors from more diverse backgrounds. To mitigate endogeneity concerns, we first employ the IV approach to check the robustness of our results.

We use the population diversity of the county where the firm is headquartered as the IV for board diversity, following the literature (e.g., Anderson et al. (2011), Davis and Henderson (2008), Fama and Jensen (1983), McPherson and Smith-Lovin (1987), and Pirinsky and Wang (2006)). Since many corporate directors are from the local geographic area of firms' headquarters, a more diverse local population makes it easier for firms to source heterogeneous corporate directors. We collect the sample firms' historical headquarters through their Securities and Exchange Commission filings, including both the 10Q and 10K forms. We follow Anderson et al. (2011) to construct the composite measure of county population diversity (CNY_DIVERSITY) based on the local population distribution of age, race, gender, and employment characteristics.²² We obtain local economic and demographic information from the U.S. Census Bureau, Bureau of Economic Analysis, and Bureau of Labor Statistics.

Table 9 reports the results of the 2-stage least squares analysis. In the first stage, we regress the board diversity index against CNY_DIVERSITY, the instrumental variable, and the same set of control variables used in column 5 of Table 2. As column 1 shows, the instrumental variable is significant at the 1% level and the Kleibergen-Paap rk Wald *F*-statistic is 45.29, higher than the 10% critical value of 16.38, indicating the instrument is not weak.

In the second stage, we regress PATENT against the fitted value of board diversity index as well as the control variables. Column 2 of Table 9 shows the estimated coefficient of the fitted value of board diversity is positive and significant at the 1% level, consistent with the ordinary least squares (OLS) regression results.

²²First, we calculate the age HHI based on the distribution of 18 age brackets in each county; the percentage of female population of the county; the race HHI based on the percentage of six ethnic groups; and the employment HHI based on the percentage of employees among the 19 North American Industry Classification System industries. We then rank the counties into quartiles by each of the four diversity measures. Finally, we sum up the four rankings of each county and divide the sum by 16 to obtain the composite county diversity index, which ranges from 0.25 to 1.

TABLE 9 Instrumental Variable Estimation

Table 9 reports the results from the 2-stage least squares regression. In the first-stage regression, the dependent variable is the board diversity index (BRD_DIVERSITY) and the instrumental variable (IV) is the county diversity index (CNV_DIVERSITY) in column 1. In the second-stage regression, the dependent variable is PATENT (column 2). Columns 3 and 4 report the results with additional variables controlling for local economics and demographic characteristics. GDP_GROWTH, EDUCATION, EMPLOYMENT, and CA_MA are as defined in the Appendix. All regressions include the industry and year fixed effects. The t-statistics (in parentheses) are based on firm-level clustered standard errors. *, **, and *** indicate statistical significance at the 10%, 5%, and 1% levels, respectively.

	BRD_DIVERSITY	PATENT _{t+3}	BRD_DIVERSITY	PATENT _{t+3}
Dependent Variable	1 1st Stage	2 2nd Stage	3 1st Stage	4 2nd Stage
BRD_DIVERSITY		3.308*** (3.25)		2.583*** (2.82)
CNY_DIVERSITY	0.035*** (6.70)		0.034*** (6.41)	
GDP_GROWTH			0.071 (1.36)	-1.007 (-1.07)
EDUCATION			0.001*** (3.80)	0.011*** (4.90)
EMPLOYMENT			0.003 (0.96)	-0.011 (-0.16)
CA_MA			0.007** (2.51)	0.215*** (7.16)
Control variables Industry effects Year effects	Yes Yes Yes	Yes Yes Yes	Yes Yes Yes	Yes Yes Yes
No. of obs. R^2	10,511 0.114	10,511 0.517	10,511 0.115	10,511 0.518
Kleibergen-Paap rk Wald F-statistic	45.29	_	41.45	_

Columns 3 and 4 of Table 9 show the IV estimation results hold after controlling for important local economic and demographic characteristics, including economic growth rate, employment rate, education level, and whether the county is located in California or Massachusetts.

B. Alternative Measures of Innovation

Thus far, we have mainly used PATENT as the measure of innovation. We now test whether the baseline regression results hold when the dependent variable is one of the five alternative innovation measures described in Section II: ADJ_PATENT, CITATION, ADJ_MV, AVG_EXT_CITATION, and ADJ_AVG_EXT_CITATION. Panel A of Table 10 shows the estimated coefficients of the board diversity index remain significantly positive for all five alternative innovation measures.

C. Controlling for CEO Influence

Given the influence of CEOs on firm innovation, we check whether our results hold after controlling for i) CEO fixed effects and ii) four distinct CEO characteristics that the literature has shown are related to corporate innovation: CEO general skills, CEO overconfidence, inventor CEOs, and the deferred compensation of CEOs (see Custodio et al. (2019), Galasso and Simcoe (2011), Hirshleifer et al. (2012), Islam and Zein (2018), and Sunder et al. (2017)). CEO_GENERAL_ABILITY measures the generality of a CEO's human capital based on the CEO's lifetime work experience in publicly traded firms prior to

TABLE 10 Robustness Checks

The dependent variable in Panel A of Table 10 is one of the five alternative measures of innovation. Panel B controls for chief executive officer (CEO) fixed effects in column 1 and four CEO characteristics in column 2. Panel C controls for additional firm and local variables that may affect innovation. All regressions control for industry and year fixed effects. The *t*-statistics (in parentheses) are based on firm-level clustered standard errors. The table also presents the number of observations and the R^2 values for each regression. *, **, and *** indicate statistical significance at the 10%, 5%, and 1% levels, respectively. All variables are defined in the Appendix.

Panel A. Alternative Measures of Innovation

Dependent Variable	ADJ_PATENT _{t+3}	ADJ_MV _{t+3}		AVG_EXT_ CITATION _{t+3}	ADJ_AVG_ EXT_ CITATION ₁₊₃
BRD_DIVERSITY	0.967** (2.22)	0.585* (1.82)	0.418** (2.52)	0.150** (2.36)	0.139** (2.27)
Control variables Industry effects Year effects	Yes Yes Yes	Yes Yes Yes	Yes Yes Yes	Yes Yes Yes	Yes Yes Yes
No. of obs. R^2	10,850 0.305	10,850 0.319	10,850 0.258	10,850 0.292	10,850 0.248
Panel B. CEO Influen	ce				
				PATENT _{t+3}	
Dependent Vari	able		1		2
BRD_DIVERSITY			0.675*** (3.09)		0.791*** (2.87)
CEO_GENERAL_ABI	LITY				0.006 (1.37)
CEO_OVERCONFIDE	INCE				0.113*** (3.83)
CEO_DEFER_COMP					0.197*** (3.51)
CEO_INVENTOR					0.058* (1.67)
Control variables CEO effects Industry effects Year effects			Yes Yes Yes Yes		Yes No Yes Yes
No. of obs. R^2			8,571 0.903		6,888 0.533

Panel C. Additional Innovation Determinants

	PATE	INI _{t+3}	
Dependent Variable	1	2	
BRD_DIVERSITY	0.740** (2.09)	0.796** (2.53)	
ANTITAKEOVER_PROVISIONS	0.023*** (2.92)	0.029** (2.25)	
ANALYST_COVERAGE	0.016* (1.68)	0.017* (1.95)	
STOCK_LIQUIDITY	17.905** (-2.46)	-16.373* (-1.90)	
INDUSTRY_HHI	-1.338 (-0.97)	-1.201 (-1.18)	
INDUSTRY_HHI ²	1.063 (1.38)	1.265 (1.54)	
EDUCATION		0.004* (1.67)	
CA_MA		0.453*** (2.83)	
Control variables Industry effects Year effects	Yes Yes Yes	Yes Yes Yes	
No. of obs. R^2	7,461 0.538	7,303 0.527	

the current CEO position;²³ CEO_OVERCONFIDENCE is a dummy variable that equals 1 if the CEO postpones the exercise of vested options that are at least 67% in the money, and 0 otherwise; CEO_DEFERRED_COMP is the ratio of the estimated dollar value of the CEO's options and other deferred compensation to the CEO's total compensation; and CEO_INVENTOR is a dummy variable that equals 1 if the CEO has filed at least one patent, and 0 otherwise. We rely on the U.S. Patent Inventor Database to identify CEO inventors. ExecuComp provides the data to construct our measures of CEO overconfidence and deferred compensation.

Panel B of Table 10 reports the results. The estimated coefficients of the board diversity index are significantly positive in column 1, where we control for CEO fixed effects, and in column 2, where we control for the four CEO characteristics.

D. Additional Robustness Checks

In this section, we test whether our results hold after controlling for additional variables. First, we control for the following variables that the literature shows are related to corporate innovation in column 1 of Panel C in Table 10: antitakeover provisions, analyst coverage, stock liquidity, and industry competition (see Aghion et al. (2005), Atanassov (2013), Chemmanur and Tian (2018), Fang et al. (2014), and He and Tian (2013)). Specifically, ANTITAKEOVER_PROVISIONS is the number of antitakeover provisions in the corporate charter and bylaws obtained from ISS (formerly RiskMetrics) as in Gompers, Ishii, and Metrick (2003): ANALYST_COVERAGE is the logarithm of 1 plus the number of 1-year-ahead analyst earnings forecasts for a firm as reported in Institutional Brokers' Estimate System (I/B/E/S); STOCK_LIQUIDITY is the negative logarithm of 1 plus the high-low spread estimator as in Corwin and Schultz (2012);²⁴ INDUSTRY_HHI is the revenue-based HHI of an industry as a measure of market competition; and INDUSTRY_HHI² is the square of the HHI. Panel C of Table 10 reports the results. In column 2, we add the local variables that are statistically significant in the IV estimation: EDUCATION and CA_MA. The estimated coefficient of board diversity remains significantly positive in both columns after controlling for these additional variables.

As a final robustness check, we re-estimate the baseline regression using firm-year observations where the number of patents is not 0, and our main results hold.

VI. Conclusion

In this article, we investigate the relation between board diversity and corporate innovation. We construct a multidimensional diversity index based on

²³We thank Claudia Custodio for providing the general ability index, which is the first factor of the principal components analysis of the five aspects of a CEO's professional career: past number of i) positions, ii) firms, and iii) industries in which a CEO worked; iv) whether the executive held a CEO position at a different company; and v) whether the CEO worked for a conglomerate firm. Please see Custodio, Ferreira, and Matos (2019) for detailed variable definition.

²⁴We put a negative sign in front of the logarithm so that an increase in spread corresponds to lower stock liquidity.

six aspects of director characteristics: education, demographic attributes, cultural attitudes, managerial traits, professional background, and board experience. We find the composite index of board diversity to be positively related to innovation output. Moreover, we compare the innovation strategies undertaken by firms with different board compositions. We find evidence to suggest that firms with diverse boards engage in more exploratory innovation, with the development of new technology in unfamiliar areas. These findings suggest that corporate innovation benefits from the strategic resources brought by directors with different backgrounds. A diverse board fosters a culture of innovation and creativity, which encourages thinking outside the box and ventures into novel technologies.

However, policy makers and shareholders should exercise caution in promoting diversity in the board rooms. Differences between directors can create communication and coordination problems in the boardroom, protracting the decisionmaking process. We find a nonlinear relation between board diversity and corporate innovation, consistent with the cost–benefit tradeoff of board diversity. Piecewise regressions reveal that the positive relation between diversity and innovation disappears as the board becomes increasingly diverse. Moreover, we find the benefits of board diversity are more pronounced for firms with complex operations, with more experienced directors, and that are subject to stronger external governance, suggesting the benefits are mainly due to the superior advising capacity of diverse boards. Finally, of the six different aspects of board diversity, professional diversity matters the most as far as innovation is concerned.

Appendix. Variable Definitions

- PATENT: Natural logarithm of 1 plus the firm's total number of patent applications filed in a year that are eventually granted.
- ADJ_PATENT: Natural logarithm of 1 plus the firm's total class-adjusted patent counts in a year, which is patent counts divided by the average number of patents of all firms in the same technology class for that year.
- CITATION: Natural logarithm of 1 plus the firm's total class-adjusted citation count in a year.
- ADJ_MV: Natural logarithm of 1 plus the adjusted market value of the firms' new patents.
- AVG_EXT_CITATION: Natural logarithm of 1 plus the average number of non-selfcitations per patent received by the firm's patents in a year.
- ADJ_AVG_EXT_CITATION: Natural logarithm of 1 plus the firm's average external citation count, scaled by the average number of external citations of all patents in the same technology class in a year.
- UNKNOWN: Natural logarithm of 1 plus the number of patents filed in previously unknown areas.
- TECH_PROXY: Technological proximity between the firm's patent portfolio in 1 year and that of the same firm in prior years.
- TOP1_CITE: Natural logarithm of 1 plus the number of patents in the top 1% of the citation distribution within the technology class and application year.
- TOP10_CITE: Natural logarithm of 1 plus the number of patents in the top 10% of the citation distribution within the technology class and application year (excluding the top 1%).

- MIDDLE_CITE: Natural logarithm of 1 plus the number of patents that are cited but are not in the top 10% of the citation distribution.
- NO_CITE: Natural logarithm of 1 plus the number of patents that receive no citations.

BRD_AGE: Natural logarithm of the average age of the board members.

BRD_FEMALE: Proportion of female directors on the board.

BRD_PHD: Proportion of board directors with PhD degrees.

BRD_FOUNDER: Proportion of directors with experience founding a business.

- BRD_TENURE: Natural logarithm of average director tenure.
- BRD_EXT_SEAT: Natural logarithm of the average number of directors' external board seats.
- BRD_INDEPENDENCE: Proportion of independent directors.

BRD_POLITICAL_CONNECTION: Proportion of politically connected directors.

BRD_MILITARY: Proportion of directors with military experience.

BRD_DEPRESSION_BABY: Proportion of directors born during the Great Depression.

POWER_DISTANCE: Average power distance scores of directors divided by 100.

- INDIVIDUALISM: Average individualism scores of directors divided by 100.
- UNCERTAINTY_AVOIDANCE: Average uncertainty avoidance scores of directors divided by 100.
- MASCULINITY: Average masculinity scores of directors divided by 100.
- LONG_TERM_ORIENTATION: Average long-term orientation scores of directors divided by 100.
- CNY_DIVERSITY: Composite measure of county population diversity based on the local population distribution of age, race, gender, and employment characteristics.
- EDUCATION: Proportion of county population with the education of high school completion or above.
- EMPLOYMENT: County employment rate.
- GDP_GROWTH: GDP growth rate during the past 3 years.
- CA_MA: Dummy variable which equals to 1 if the firm headquarter is located in either California or Massachusetts, and 0 otherwise.
- CEO_GENERAL_ABILITY: Generality of a CEO's human capital based on the CEO's lifetime work experience in publicly traded firms prior to the current CEO position.
- CEO_OVERCONFIDENCE: Dummy variable that equals 1 if the CEO postpones the exercise of vested options that are at least 67% in the money, and 0 otherwise.
- CEO_DEFERRED_COMP: Ratio of the estimated dollar value of the CEO's options and other deferred compensation to the CEO's total compensation.
- CEO_INVENTOR: Dummy variable that equals 1 if the CEO has filed at least one patent, and 0 otherwise.
- ANTITAKEOVER_PROVISIONS: Number of antitakeover provisions in the corporate charter and bylaws.
- ANALYST_COVERAGE: Natural logarithm of 1 plus the number of 1-year-ahead analyst earnings forecasts for a firm.
- STOCK_LIQUIDITY: Negative natural logarithm of 1 plus the high-low spread estimator.
- INDUSTRY_HHI: Revenue-based HHI of an industry as a measure of market competition; and
- INDUSTRY_HHI²: Square of INDUSTRY HHI.
- SIZE: Natural logarithm of total assets (in \$millions).

- AGE: Natural logarithm of 1 plus the number of years since the firm's first appearance in the Center for Research in Security Prices (CRSP).
- LEV: Total debt divided by total assets.
- ROA: Operating income before depreciation divided by total assets.
- CASH: Cash holdings divided by total assets.
- CF: Sum of income before extraordinary items and depreciation divided by total assets.
- TOBINS_Q: (Assets + year-end closing price \times year-end number of outstanding shares book equity)/assets.
- TANGIBLE: Property, plants, and equipment divided by total assets.
- R&D: R&D expenditures divided by total assets.
- TURNOVER: Average ratio of the monthly trading volume to the number of shares outstanding over the fiscal year.

RISK: Natural logarithm of the standard deviation of stock daily returns in the fiscal year.

IO: Percentage of institutional ownership.

Supplementary Material

Supplementary Material for this article is available at https://doi.org/10.1017/S0022109019001005.

References

- Adams, R., and D. Ferreira. "Women in the Boardroom and Their Impact on Governance and Performance." Journal of Financial Economics, 94 (2009), 291–309.
- Adams, R., and P. Funk. "Beyond the Glass Ceiling: Does Gender Matter?" Management Science, 58 (2012), 219–235.
- Aghion, P.; N. Bloom; R. Blundell; R. Griffith; and P. Howitt. "Competition and Innovation: An Inverted-U Relationship." *Quarterly Journal of Economics*, 120 (2005), 701–728.
- Aghion, P.; J. Van Reenen; and L. Zingales. "Innovation and Institutional Ownership." American Economic Review, 103 (2013), 277–304.
- Agrawal, A., and C. R. Knoeber. "Managerial Compensation and the Threat of Takeover." Journal of Financial Economics, 47 (1998), 219–239.
- Anderson, R.; D. Reeb; A. Upadhyay; and W. Zhao. "The Economics of Director Heterogeneity." *Financial Management*, 40 (2011), 5–38.
- Atanassov, J. "Do Hostile Takeovers Stifle Innovation? Evidence from Antitakeover Legislation and Corporate Patenting." *Journal of Finance*, 68 (2013), 1097–1131.
- Balsmeier, B.; L. Fleming; and G. Manso. "Independent Boards and Innovation." Journal of Financial Economics, 123 (2017), 536–557.
- Baranchuk, N.; R. Kieschnick; and R. Moussawi. "Motivating Innovation in Newly Public Firms." Journal of Financial Economics, 111 (2014), 578–588.
- Bebchuk, L., and A. Cohen. "Firms' Decisions Where to Incorporate." *Journal of Law and Economics*, 46 (2003), 383–425.
- Belenzon, S., and T. Berkovitz. "Innovation in Business Groups." *Management Science*, 56 (2010), 519–535.
- Bernstein, S. "Does Going Public Affect Innovation?" Journal of Finance, 70 (2015), 1365-1403.
- Brickley, J.; R. Lease; and C. Smith. "Ownership Structure and Voting on Antitakeover Amendments." Journal of Financial Economics, 20 (1988), 267–291.
- Carter, D.; F. D'Souza; B. Simkins; and W. Simpson. "The Gender and Ethnic Diversity of US Boards and Board Committees and Firm Financial Performance." *Corporate Governance: An International Review*, 18 (2010), 396–414.
- Chemmanur, T. J.; E. Loutskina; and X. Tian. "Corporate Venture Capital, Value Creation, and Innovation." *Review of Financial Studies*, 27 (2014), 2434–2473.
- Chemmanur, T. J., and X. Tian. "Do Antitakeover Provisions Spur Corporate Innovation? A Regression Discontinuity Analysis." *Journal of Financial and Quantitative Analysis*, 53 (2018), 1163–1194.

- Chen, C., and T. Zhang. "Is Diversity Good? Analysis of Board Diversity and Firm Value." Working Paper, University of Dayton (2014).
- Chen, J.; W. Leung; and K. Evans. "Board Gender Diversity, Innovation and Firm Performance." Journal of Empirical Finance, 48 (2018), 236–254.
- Chen, X.; J. Harford; and K. Li. "Monitoring: Which Institutions Matter?" Journal of Financial Economics, 86 (2007), 279–305.
- Cornaggia, J.; Y. Mao; X. Tian; and B. Wolfe. "Does Banking Competition Affect Innovation?" Journal of Financial Economics, 115 (2015), 189–209.
- Cornett, M.; A. Marcus; A. Saunders; and H. Tehranian. "The Impact of Institutional Ownership on Corporate Operating Performance." Journal of Banking and Finance, 31 (2007), 1771–1794.
- Corwin, S., and P. Schultz. "A Simple Way to Estimate Bid-Ask Spreads from Daily High and Low Prices." *Journal of Finance*, 67 (2012), 719–760.
- Custodio, C.; M. Ferreira; and P. Matos. "Do General Managerial Skills Spur Innovation?" Management Science, 65 (2019), 459–476.
- Davis, J., and J. Henderson. "The Agglomeration of Headquarters." *Regional Science and Urban Economics*, 38 (2008), 445–460.
- Faleye, O.; T. Kovacs; and A. Venkateswaran. "Do Better-Connected CEOs Innovate More?" Journal of Financial and Quantitative Analysis, 49 (2014), 1201–1225.
- Fama, E., and M. Jensen. "Separation of Ownership and Control." Journal of Law and Economics, 26 (1983), 301–325.
- Fang, V. W.; X. Tian; and S. Tice. "Does Stock Liquidity Enhance or Impede Firm Innovation?" Journal of Finance, 69 (2014), 2085–2125.
- Ferreira, D. "Board Diversity." In Corporate Governance: A Synthesis of Theory, Research, and Practice, R. Anderson and H. K. Baker, eds. Hoboken, NJ: John Wiley & Sons (2010), 225–242.
- Galasso, A., and T. S. Simcoe. "CEO Overconfidence and Innovation." *Management Science*, 57 (2011), 1469–1484.
- Galia, F., and E. Zenou. "Board Composition and Forms of Innovation: Does Diversity Make a Difference?" European Journal of International Management, 6 (2012), 630–650.
- Garcia-Meca, E.; I. Garcia-Sanchez; and J. Martinez-Ferrero. "Board Diversity and Its Effects on Bank Performance: An International Analysis." *Journal of Banking and Finance*, 53 (2015), 202–214.
- Giannetti, M.; G. Liao; and X. Yu. "The Brain Gain of Corporate Boards: Evidence from China." Journal of Finance, 70 (2015), 1629–1682.
- Giroud, X., and H. M. Mueller. "Corporate Governance, Product Market Competition, and Equity Prices." Journal of Finance, 66 (2011), 563–600.
- Goldman, E.; J. Rocholl; and J. So. "Do Politically Connected Boards Affect Firm Value?" Review of Financial Studies, 22 (2009), 2331–2360.
- Gompers, P. A.; J. Ishii; and A. Metrick. "Corporate Governance and Equity Prices." *Quarterly Journal of Economics*, 118 (2003), 107–155.
- Haan, M., and Y. Riyanto. "The Effects of Takeover Threats on Shareholders and Firm Value." Journal of Economic Behavior and Organization, 59 (2006), 45–68.
- Hall, B.; A. Jaffe; and M. Trajtenberg. "The NBER Patent Citation Data File: Lessons, Insights and Methodological Tools." NBER Working Paper 8498 (2001).
- Hall, B.; A. Jaffe; and M. Trajtenberg. "Market Value and Patent Citations." RAND Journal of Economics, 36 (2005), 16–38.
- Hambrick, D., and P. Mason. "Upper Echelons: The Organization as a Reflection of Its Top Managers." Academy of Management Review, 9 (1984), 193–206.
- He, J., and X. Tian. "The Dark Side of Analyst Coverage: The Case of Innovation." Journal of Financial Economics, 109 (2013), 856–878.
- He, J., and X. Tian. "Finance and Corporate Innovation: A Survey." Asia-Pacific Journal of Financial Studies, 47 (2018), 165–212.
- Hirshleifer, D.; A. Low; and S. Teoh. "Are Overconfident CEOs Better Innovators?" Journal of Finance, 67 (2012), 1457–1498.
- Hofstede, G. Culture's Consequences: International Differences in Work-Related Values. Beverly Hills, CA: Sage Publications (1984).
- Islam, M., and J. Zein. "Inventor CEOs." Working Paper, American Economic Association 2018 Annual Conference (2018).
- Jaffe, A. B. "Characterizing the 'Technological Position' of Firms, with Application to Quantifying Technological Opportunity and Research Spillovers." *Research Policy*, 18 (1989), 87–97.
- Kogan, L.; D. Papanikolaou; A. Seru; and N. Stoffman. "Technological Innovation, Resource Allocation, and Growth." *Quarterly Journal of Economics*, 132 (2017), 665–712.
- Lerner, J.; M. Sorensen; and P. Strömberg. "Private Equity and Long-Run Investment: The Case of Innovation." *Journal of Finance*, 66, 2 (2011), 445–477.

182 Journal of Financial and Quantitative Analysis

- Levine, J., and R. Moreland. "Group Processes." In Advanced Social Psychology, A. Tesser, ed. New York, NY: McGraw-Hill (1994), 419–465.
- Malmendier, U.; G. Tate; and J. Yan. "Overconfidence and Early-Life Experiences: The Effect of Managerial Traits on Corporate Financial Policies." *Journal of Finance*, 66 (2011), 1687–1733.
- Manzoni, J.-F.; P. Strebel; and J.-L. Barsoux. "Why Diversity Can Backfire on Company Boards." Wall Street Journal, (2010), R3.
- McPherson, J., and L. Smith-Lovin. "Homophily in Voluntary Organizations: Status Distance and the Composition of Face-to-Face Groups." *American Sociological Review*, 52 (1987), 370–379.
- Meulbroek, L.; M. Litchell; H. Mulherin; J. Netter; and A. Poulsen. "Shark Repellents and Managerial Myopia: An Empirical Test." *Journal of Political Economy*, 98 (1990), 1108–1117.
- Midavaine, J.; W. Dolfsma; and R. Aalbers. "Board Diversity and R&D Investment." Management Decision, 54 (2016), 558–569.
- Nielsen, S., and M. Huse. "The Contribution of Women on Boards of Directors: Going beyond the Surface." Corporate Governance: An International Review, 18 (2010), 136–148.
- Pirinsky, C., and Q. Wang. "Does Corporate Headquarters Location Matter for Stock Returns?" Journal of Finance, 61 (2006), 1991–2015.
- Porter, M. E. "Capital Disadvantage: America's Failing Capital Investment System." Harvard Business Review, 70 (1992), 65–82.
- Rhode, D., and A. Packel. "Diversity on Corporate Boards: How Much Difference Does Difference Make?" *Delaware Journal of Corporate Law*, 39 (2014), 377–426.
- Rosenberg, S.; D. Erlick; and L. Berkowitz. "Some Effects of Varying Combinations of Group Members on Group Performance Measures and Leadership Behaviors." *Journal of Abnormal and Social Psychology*, 51 (1955), 195–203.
- Schilling, M. A., and C. C. Phelps. "Interfirm Collaboration Networks: The Impact of Large-Scale Network Structure on Firm Innovation." *Management Science*, 53 (2007), 1113–1126.
- Seru, A. "Firm Boundaries Matter: Evidence from Conglomerates and R&D Activity." Journal of Financial Economics, 111 (2014), 381–405.
- Steiner, I. Group Process and Productivity. New York, NY: Academic Press (1972).
- Sunder, J.; S. V. Sunder; and J. Zhang. "Pilot CEOs and Corporate Innovation." Journal of Financial Economics, 123 (2017), 209–224.
- Teachman, J. "Analysis of Population Diversity." Sociological Methods and Research, 8 (1980), 341–362.
- Torchia, M.; A. Calabro; and M. Huse. "Women Directors on Corporate Boards: From Tokenism to Critical Mass." *Journal of Business Ethics*, 102 (2011), 299–317.
- Triana, M.; T. L. Miller; and T. M. Trzebiatowski. "The Double Edged Nature of Board Gender Diversity: Diversity, Firm Performance, and the Power of Women Directors as Predictors of Strategic Change." Organization Science, 25 (2014), 609–632.
- Tuckman, B. "Personality Structure, Group Composition, and Group Functioning." Sociometry, 27 (1964), 469–487.
- Upadhyay, A., and H. Zeng. "Gender and Ethnic Diversity on Boards and Corporate Information Environment." *Journal of Business Research*, 67 (2014), 2456–2463.
- Watson, W. E.; K. Kumar; and L. K. Michaelsen. "Cultural Diversity's Impact on Interaction Process and Performance: Comparing Homogeneous and Diverse Task Groups." *Academy of Management Journal*, 36 (1993), 590–602.
- Wiersema, M. F., and K. Bantel. "Top Management Demography and Corporate Strategic Change." Academy of Management Journal, 35 (1992), 91–121.